Author's response to reviews

Title: Effectiveness of a structured education reminiscence-based programme for staff on the quality of life of residents with dementia in long-stay units: A study protocol for a cluster randomised trial

Authors:

Eamon O'Shea (eamon.oshea@nuigalway.ie)
Declan Devane (declan.devane@nuigalway.ie)
Kathy Murphy (kathy.murphy@nuigalway.ie)
Adeline Cooney (adeline.cooney@nuigalway.ie)
Dymphna Casey (dymphna.casey@nuigalway.ie)
Fionnuala Jordan (fionnuala.jordan@nuigalway.ie)
Andrew Hunter (andrew.hunter@nuigalway.ie)
Edel Murphy (edel.murphy@nuigalway.ie)

Version: 2 Date: 28 January 2011

Author's response to reviews:

The following changes were made to the paper:

• Changed the title as requested
• Page 3, Background, 3rd para, 10th line: changed “psychological” to “psychosocial”.
• Page 5, Methods/Design: second line: changed reference to Figure 1 to: (see Figure 1).
• Page 5, Methods/Design: last line: removed caption to Figure 1.
• Page 10, Analyses: Quantitative Data: 4th paragraph: drop reference to demographic characteristics of units.
• Page 14: added the caption for Figure 1
• A small number of minor grammatical/typographical changes (commas etc.) made

Please let me know if you require any further detail.